Author's response to reviews

Title: The Effectiveness of Video Interaction Guidance in Parents of Premature Infants: a Multicenter Randomised Controlled Trial

Authors:

Anneke Tooten (a.tooten@uvt.nl)
Hannah N. Hoffenkamp (h.n.hoffenkamp@uvt.nl)
Ruby A.S. Hall (r.a.s.hall@uvt.nl)
Frans Willem Winkel (f.w.winkel@uvt.nl)
Marij Eliëns (m.eliens@combinatiejeugdzorg.nl)
Ad J.J.M. Vingerhoets (vingerhoets@uvt.nl)
Hedwig J.A. van Bakel (h.j.a.vanbakel@uvt.nl)

Version: 2 Date: 10 November 2011

Author's response to reviews:

Tilburg, November 10th 2011

Dear editorial board of BMC Pediatrics,

Thank you very much for your consideration of our manuscript. We have enlisted all points to consider below, and address each one individually. As you will see, we have made every attempt to incorporate these suggestions as thoroughly as possible.

- This study is externally funded by a Dutch charitable foundation: “Achmea, Foundation Victim and Society” (Stichting Achmea Slachtoffer en Samenleving), Project number: NL24021.060.08. Evidence of this approval can be found in the first enclosed attachment, including details of the amount of funding (EUR 768.875).

- Medical ethical approval was obtained from Catharina Hospital, Eindhoven, the Netherlands on 21-10-2008. Evidence of this approval can be found in the second enclosed attachment.

(Unfortunately both documents are only available in Dutch. If you have any questions concerning the contents, we will be more than willing to be of assistance)

- In the revised manuscript (10-11-2011) you can find the changes made in the section ‘Acknowledgements’.

- Currently there are no other manuscripts under submission with another journal based on this study protocol.

Hopefully the above mentioned additional information and documents are in compliance with your request.
Sincerely,

Anneke Tooten, on behalf of the authors

Corresponding author: A. Tooten, International Victimology Institute Tilburg, Tilburg University, The Netherlands. E-mail address: a.tooten@uvt.nl, phone number: (+31) 13 4668759.